Title: Sinus versus nonsinus tachycardia in the emergency department: importance of age and heart rate

Authors:
Duane S Pinto (dpinto@caregroup.harvard.edu)
Kalon K Ho (kho@hcri.harvard.edu)
Peter J Zimetbaum (pzimetba@caregroup.harvard.edu)
Alexander Pedan (alex@clinepi.bwh.harvard.edu)
Ary L Goldberger (agoldber@caregroup.harvard.edu)

Version: 4 Date: 31 Jul 2003

Single Capitalization typo fixed in Title.